Author's response to reviews

Title: CAM therapies among primary care patients using opioid therapy for chronic pain

Authors:

Sara Fleming (sara.fleming@bastyr.edu)
David P Rabago (drabago@fammed.wisc.edu)
Marlon P Mundt (mmundt@fammed.wisc.edu)
Michael F. Fleming (mfleming@fammed.wisc.edu)

Version: 5 Date: 2 March 2007

Author's response to reviews:

Dear Biomed,
We submitted a revised manuscript (info below) on Feb 28. We have found 1 error in the manuscript that needs to be changed before publishing it.

I have uploaded a new version on the website. Specifically the things changed were:
Abstract: Results section: 2nd line: duration of pain was changed from "7" to "16" years.
In the Methods section: paragraph 3: 4th line: duration of pain was changed from "7.1" to "16"

We apologize for making a change so late, but we want to ensure that our data is correct.

CAM therapies among primary care patients using opioid therapy for chronic pain
journal BMC Complement Altern Med
manuscript ID 1867713472112502
submitted 16 Aug 2006
revised 28 Feb 2007
peer review status Revision with reviewers